Author's response to reviews

Title: Clinical significance of retinal emboli during diagnostic and therapeutic cardiac catheterization in patient with coronary artery disease

Authors:

javad kojuri (kojurij@yahoo.com)
morteza mehdizadeh (mehdizadeh@sums.ac.ir)
hamedrostami (rostam56@gmail.com)
danial shahidian (shahidd@yahoo.com)

Version: 4 Date: 17 December 2010

Author's response to reviews:

Dear Editors:

Thank you so much for accepting our article. We added the competing interest and Authors' contributions sections to the MS. We formatted the article based on BMC format. If there is more changes needed let us know.

Sincerely Kojuri J corresponding author